Author's response to reviews

Title: A comparison between the APACHE II and Charlson Index Score for predicting hospital mortality in critically ill patients: a population based cohort study

Authors:

Susan Quach (susan.quach@phac-aspc.gc.ca)
Deirdre A Hennessy (dhenness@ucalgary.ca)
Andrew Fong (andrew.fong@calgaryhealthregion.ca)
Peter Faris (faris@ucalgary.ca)
Hude Quan (hquan@ucalgary.ca)
Christopher Doig (Chip.Doig@calgaryhealthregion.ca)

Version: 3 Date: 26 March 2009

Author's response to reviews: see over
March 26th 2009

Melissa Norton, MD
Editor-in-Chief
BMC Health Services Research

Re: A comparison between the APACHE II and Charlson Index Score for predicting hospital mortality in critically ill patients: a population based cohort study- Response to editorial concern

Dear Dr. Norton,

Thank you for conditionally accepting our manuscript for publication in BMC Health Services Research. We have added an acknowledgments section to the manuscript, as required by your editorial staff. The funding body from which Ms. Susan Quach received support did not provide any funding for manuscript preparation or submission, was not involved in study design, data collection, data analysis, interpretation of data or in the writing of the manuscript.

Thanks and regards
Yours sincerely,

Christopher James Doig MD MSc FRCPC
Associate Professor, Faculty of Medicine, University of Calgary
Medical Director, Foothills Multisystem Intensive Care Unit